Author’s response to reviews

Title: Back to past leeches: Repeated phlebotomies and cardiovascular risk

Authors:

Melania Manco (melaniamanco@tiscali.it)
José Manuel Fernandez Real (uden.jmfernandezreal@htrueta.scs.es)

Version: 4 Date: 16 February 2012

Author’s response to reviews:

Dear Editors,

We would like to express you our gratitude for editing our manuscript and suggestions which have contributed to the generation of a better editorial.

Please, find enclosed our point-by-point responses along with the copy of the revised manuscript. Virtually all comments and suggestion made by you have been addressed in our revised manuscript.

Sincerely Yours,
Melania Manco

REPLY TO EDITORS’ COMMENTS
E-mail addresses have been provided for both authors
A short introduction on the metabolic syndrome has been inserted in the text.
We detailed how findings from the study may advance the field as suggested in the first part of the main text and in the conclusion.
Authors’ contributions, authors’ information and acknowledgements have been included.
References have been formatted in the journal style.